23335084



Earth and Space Science

RESEARCH ARTICLE

10.1029/2022EA002664

Key Points:

- We show high spatial and temporal variability of CO₂ along 40 alkaline fluvial networks in Denmark
- In summer, the CO₂ decreases markedly upon transit through lakes, outlets having mean concentrations close to air saturation values
- CO₂ concentrations average 9.2-fold supersaturation at sites not influenced by lakes and daily mean emission for streams was 2.1 g CO₂-C m⁻²

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

K. T. Martinsen, kenneth.martinsen@bio.ku.dk

Citation:

Sand-Jensen, K., Riis, T., Kjær, J. E., & Martinsen, K. T. (2022). Stream-lake connectivity is an important control of fluvial CO₂ concentrations and emissions in catchments. *Earth and Space Science*, *9*, e2022EA002664. https://doi. org/10.1029/2022EA002664

Received 5 OCT 2022 Accepted 8 NOV 2022

Author Contributions:

Conceptualization: Kaj Sand-Jensen Data curation: Kaj Sand-Jensen, Tenna Riis Formal analysis: Kaj Sand-Jensen, Kenneth Thorø Martinsen

Funding acquisition: Kaj Sand-Jensen, Tenna Riis Investigation: Kaj Sand-Jensen, Tenna Riis

Methodology: Kaj Sand-Jensen, Tenna Riis

Software: Kenneth Thorø Martinsen Visualization: Johan Emil Kjær, Kenneth Thorø Martinsen

© 2022 The Authors. Earth and Space Science published by Wiley Periodicals LLC on behalf of American Geophysical Union.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Stream-Lake Connectivity Is an Important Control of Fluvial CO₂ Concentrations and Emissions in Catchments

Kaj Sand-Jensen¹ , Tenna Riis² , Johan Emil Kjær¹ , and Kenneth Thorø Martinsen¹

¹Freshwater Biological Laboratory, Department of Biology, University of Copenhagen, Copenhagen, Denmark, ²Institute of Biology, Aarhus University, Aarhus C, Denmark

Abstract Streams in cultivated lowlands are commonly supersaturated with CO_2 and are a source of CO_2 to the atmosphere. Great uncertainties exist regarding the spatiotemporal variations of CO_2 concentrations and emission rates in stream-lake fluvial networks and small streams with variable plant cover. We studied this variability and the underlying mechanisms in 40 small, high-alkalinity Danish streams, including 5 catchments with lakes. Generally CO_2 concentrations were, on average, 9.2 times those of the atmosphere, declining downstream with rising water temperature, chlorophyll *a* concentration, and decreasing groundwater inputs. We furthermore observed that the concentrations of CO_2 in stream waters declined at the outlet of lakes to values close to or below air saturation due to phytoplankton uptake and atmospheric loss during the long water retention time in the lakes. Downstream, CO_2 concentrations were observed to decrease in summer and in the afternoons, which indicate plant uptake of CO_2 . Sites with deeper water and few plants and low gas transfer velocity retained high CO_2 concentrations. Among 38 fluvial networks where emission could be calculated, it varied 10-fold (0.41–4.06 g C m⁻² d⁻¹), but the overall mean was constrained to a narrow confidence interval (1.75–2.50 g C m⁻² d⁻¹). Our results highlight that a complex of physical, chemical and biological processes cause highly variable carbon dynamics and CO_2 emissions in fluvial networks at local and catchment scales making upscaling challenging.

Plain Language Summary Freshwater streams are supersaturated in CO_2 and release high quantities of this gas into the atmosphere. We studied the spatial and temporal variation of CO_2 concentrations in 40 fluvial networks in Denmark; of these, five include lakes. We found that the concentration of CO_2 in streams is, on average, 9.2 times that of the atmosphere. However, CO_2 concentrations decrease (a) downstream of lakes, (b) in summer, and (c) in the afternoon. These spatial and temporal trends suggest that lakes and stream plants play a significant role in regulating the concentration of CO_2 in freshwater networks. Based on observations that CO_2 concentrations measured at the outlet of eutrophic lakes may vary up to 100 times between summer, when phytoplankton blooms, and winter, we conclude that biological activity strongly influence CO_2 freshwater uptake and emission in catchments.

1. Introduction

Small lowland streams in open agricultural landscapes are highly dynamic ecosystems due to variable hydrology, chemistry, and biology, both within and among streams (Borges et al., 2018; Fuß et al., 2017; Sand-Jensen & Staehr, 2012). These streams are usually supersaturated with carbon dioxide (CO_2) and release large quantities of CO_2 into the atmosphere (Cole et al., 2007; Raymond et al., 2013). Recurring revisions of carbon budget estimates reinforce and extend the global relevance of this source of carbon emissions (Drake et al., 2018). The surplus of CO_2 in streams is supplied partly by drainage water from land and partly by internal respiratory processing of terrestrially derived organic carbon (Crawford et al., 2016; Hotchkiss et al., 2015; Marx et al., 2017). This combination often leads to high emission rates from stream reaches (Wallin et al., 2011), especially when the gas transfer velocity is high and photosynthetic CO_2 consumption is low (Sand-Jensen & Staehr, 2012). The delivery of CO_2 from a catchment to associated streams varies at sub-daily to annual scales, due to variable CO_2 release by soil respiration, CO_2 consumption by weathering of carbonate and aluminum-silicate minerals, and changing runoff (Berner & Berner, 2012; Johnson et al., 2008; Maberly et al., 2015; Rebsdorf et al., 1991). Once water-borne, CO_2 distribution is influenced by photosynthetic and respiratory processes that change with light and water temperature during the day and night and with advective and diffusive export rates (Gómez-Gerner, 2021; Sand-Jensen & Staehr, 2012). Thus, alterations of catchment delivery of



Writing – original draft: Kaj Sand-Jensen Writing – review & editing: Kaj Sand-Jensen, Tenna Riis, Johan Emil Kjær, Kenneth Thorø Martinsen CO_2 and in-stream balances of photosynthesis and respiration may cause profound changes in CO_2 concentrations and emission rates along the stream (Crawford et al., 2016; Duvert et al., 2018).

Due to the unidirectional water flow and the relative decrease of terrestrial input through the fluvial network, the relative influence of metabolism increases downstream (Hotchkiss et al., 2015). In combination with the physical gas exchange at the air-water interface, this may push CO_2 concentrations toward air saturation and, consequently, emissions to lower levels. However, the decrease in CO_2 concentration downstream in small headwater streams characterized by high CO_2 influx from land and high emission rates to the atmosphere (Finlay, 2003; Liu & Raymond, 2018) may diverge from expectations if the water transits through reaches with particularly high or low rates of CO_2 supply or loss through in-stream processes (Maurice et al., 2017). Lake outlets may show particularly low CO_2 concentrations and emissions due to atmospheric and photosynthetic CO_2 losses during extended water retention time in the lakes (Brinkerhoff et al., 2021; Sand-Jensen & Frost-Christensen, 1998). Thus, lakes may strongly influence the water–atmosphere carbon dynamics, by lowering CO_2 concentrations and emissions in the streams.

To estimate CO_2 emissions at large spatial scales, stream order categories have been used as aggregation units (Raymond et al., 2013). Other recent studies use catchment characteristics to interpolate CO_2 concentrations in the stream (Martinsen et al., 2020a). Lauerwald et al. (2015) produced high-resolution global maps of CO_2 concentrations and CO_2 emission, but excluded the smallest streams. Among other variables, Lauerwald et al. (2015) identified average catchment slope as an important predictor of CO_2 concentration. Catchment slope may influence changes in hydrological CO_2 input and it may influence atmospheric output due to the coupling between catchment slope and flow velocity with gas transfer velocity (Hutchins et al., 2019; Smits et al., 2017).

Our study is different because it includes short, small streams and mixed fluvial networks of streams and lakes, which characterize most agricultural lowland countries such as Denmark. The elevation is modest (<185 m), 75% of stream reaches are less than 2.5 m wide and they may support dense plant beds because of their shallowness (Sand-Jensen et al., 2006). About 30% of the Danish fluvial networks include eutrophic lakes that may reduce downstream CO₂ concentrations and emissions (Sand-Jensen, Riis, & Martinsen, 2022).

In this study, we present and discuss data for CO_2 concentration, water temperature, and flow velocity in 40 Danish streams, acquired in 1997–1998 as part of an extensive biodiversity campaign, complemented by biweekly measurements conducted from August 1995 to August 1996 in 18 sites along four streams that include lakes. We use an empirical relationship between flow velocity and gas transfer velocity (Sand-Jensen & Staehr, 2012) to calculate the stream-air CO_2 flux.

We show that: (a) CO_2 concentration and emission decrease along the main course of streams owing to increased stream dimension, water retention time, and water temperature and (b) CO_2 concentrations decrease extensively when water flows through phytoplankton-rich lakes.

2. Materials and Methods

2.1. Study Background

This study was originally part of an extensive analysis of plant species distribution and biodiversity conducted in 1997–1998 to evaluate local environmental conditions. The comprehensive measurements of inorganic carbon concentrations have not been previously analyzed, but they remain highly relevant even after 25 years, since no significant changes in hydrology and environmental conditions have occurred (Sand-Jensen et al., 2006). The atmospheric CO_2 concentration has increased from about 360 ppm in 1995 to 410 ppm in 2020 leading to a modest increase in the equilibrium concentration in the stream water from 15.8 to 18.0 μ M, several-fold below the common CO_2 supersaturation (159 μ M). Also, the temperature increase has been very modest (Rubek et al., 2020).

To establish the spatial differences in CO_2 concentrations and emissions, we examined 204 sites in 40 streams, 5 streams with lake influence, distributed between Denmark's three main geographical regions: Zealand (7 streams), Funen (4), and Jutland (29; Figure 1 and Table S1 in Supporting Information S1). Most streams are located in catchments with a mean annual runoff of 8–12 L s⁻¹ km⁻² for the 30 years period 1970–2000, though River Funder has a specific runoff exceeding 20 L s⁻¹ km⁻² catchment (Sand-Jensen et al., 2006). The largest catchment area (2,649 km²) drains into the River Guden, while all other streams have catchment areas <611 km²





Figure 1. Map of Denmark with the 40 studied streams along the downstream course, and their corresponding, color-coded catchments in three regions. Stream outlets are marked by solid points.

(mean 226 km²; Table S1 in Supporting Information S1). River Guden's main course was divided into an upper part with no lake influence and a lower part with prominent lake influence. We examined the downstream pattern for streams with sufficient number of examined sites (4–10) along the stream's main course (Table S2 in Supporting Information S1). Mean CO₂ emissions was calculated for 35 streams where flow velocity was available for all sites.

2.2. Sampling

All 204 sites were visited at least once in the summers of 1997 and 1998. Moreover, 18 sites in four streams on Zealand were visited biweekly the year before (August 1995-August 1996). To evaluate the suitability of using morning measurements (8:00–11:00) to represent the mean daily CO₂ concentration, the 18 sites were sampled five times during a diel cycle in July. This showed that calculated CO₂ concentrations at 9:30 (i.e., the middle of the time window 8:00–11:00), were only 8.7% (±5.7%, 95% CI) above the mean daily CO₂ concentration and, thus, our sampling approach was suitable to represent mean daily values at all 204 sites.

In each of the 204 stream sites we measured the wetted cross-sectional area, in short, wetted area (A, m^2), flow velocity, and discharge. Along a 50 m reach at each site, we created 10 transversal sections and recorded water depth at 0.2 m intervals from shore to shore to determine A. Mean flow velocity (U, m s⁻¹) over the reach was measured by adding a pulse of dissolved NaCl above the reach and recording the transit time (T, s) of 50% of the added salt below the reach from the elevated specific conductivity above the background, measured every second ($U = 50 \text{ m} \cdot T^{-1}$). Water discharge was

determined as the product $U \cdot A$ (m³ s⁻¹). At a few sites characterized by very high discharge, the salt method was not applicable; thus, flow velocity, gas transfer velocity and CO₂ emission were not calculated for these sites (Tables S1 and S2 in Supporting Information S1). Across the 204 sites, stream width averaged 6.95 m (median 5.2 m, range 0.6–30.0 m) and discharge averaged 2.78 m³ s⁻¹ (median 0.63 m³ s⁻¹, range 0.013–26.0 m³ s⁻¹).

At each site we measured water temperature and collected water samples in the morning (8:00-11:00) for measurements of pH and inorganic carbon in airtight glass bottles, which were kept cool until measurements at 20°C could be conducted in the laboratory (typically within a few hours). Chlorophyll *a* concentrations were measured in water samples from all sites in the River Guden and below lakes in other streams (Table S1 in Supporting Information S1). Chlorophyll *a* is present in planktonic and suspended benthic microalgae. Chlorophyll *a* concentrations below lakes is a measure of phytoplankton biomass and primary production in the lakes.

2.3. Water Analyses and CO₂ Concentrations and Emissions

In the laboratory, conductivity and pH were measured by standard techniques. Chlorophyll *a* concentrations in the water was measured by passing water through glass microfibre filters, extracting the filters in ethanol and measuring chlorophyll absorbance by spectrophotometry (Jespersen & Christoffersen, 1987). Total alkalinity (Acid Neutralizing Capacity, ANC), dissolved inorganic carbon pool (DIC), and CO₂ were measured by Gran-titration (Gran, 1952). In 500 measurements at the stream sites, alkalinity was consistently high (mean 2.82 mEq. L⁻¹; range 0.13–6.42 mEq. L⁻¹) and only 23 measurements were below 1.0 mEq. L⁻¹ and of those 3 were also below 0.5 mEq. L⁻¹. The minimum pH was 6.45 and only 3 measurements were below 7.0 (mean 7.6). No streams of low alkalinity and high content of dissolved organic carbon were included. Our calculation of CO₂ concentrations from alkalinity, pH, water temperature, and conductivity is very reliable because the interference of organic and mineral acids is negligible in waters of high alkalinity (>1 mEq. L⁻¹) and high pH (Abril et al., 2015; Rebsdorf et al., 1991). We confirmed this by showing close correspondence between direct DIC measurements and DIC calculated from Gran-titration (Table S3 in Supporting Information S1).

The water-air CO_2 flux (F, mmol m⁻² h⁻¹) at the field sites was calculated as the product of the water-air gradient (mmol m⁻³) and the gas transfer velocity K (m h⁻¹):

$$F = K \cdot (\mathrm{CO}_2 - \mathrm{CO}_{2-\mathrm{sat}}) \tag{1}$$

 CO_2 is the calculated concentration in the water and $CO_{2-\text{sat}}$ is the CO_2 concentration in water at air saturation at the atmospheric CO_2 partial pressure and water temperature. *K* was calculated from *U* and water temperature according to the empirical relationship established from 55 mean daily values at different sites in two Danish lowland streams (Sand-Jensen & Staehr, 2012). *K* at 20°C (K_{20} , cm h⁻¹) was calculated from the relationship (SE in parenthesis):

$$\log(K_{20}) = 1.256(\pm 0.041) + \log(U) \cdot 0.0742(\pm 0.055)$$
⁽²⁾

K at ambient water temperature was calculated from K_{20} following Thyssen and Erlandsen (1987):

$$K = K_{20} \cdot \left(1.024^{\text{temp}-20}\right) \tag{3}$$

2.4. Groundwater and Catchment Delineation

Compared to inflow of surface and drainage water, groundwater inflow is much more constant over time, so the ratio of the annual median monthly minimum discharge to the annual median discharge is a suitable index of the groundwater contribution in Danish streams (Ovesen et al., 2000). We calculated this index for our catchments where daily discharge over 50 years (1970–2020) were available from national monitoring data (MFM & DCE, 2021).

Stream catchments boundaries were derived from the national digital elevation model (DEM) at 10 m resolution (SDFE, 2021). To condition the DEM for flow routing, the DEM was preprocessed using algorithms from the RichDEM Python library (Barnes, 2018); depressions and pits in the DEM were breached (Lindsay, 2016) followed by flat resolution, where a gradient is enforced on flat surfaces to ensure flow across the DEM (Barnes et al., 2014; Tarboton, 2017). Finally, D8 flow routing (O'Callaghan & Mark, 1984) and catchment delineation were performed using the TauDEM library (Tarboton, 2017). Spatial data analysis also used GDAL (GDAL/OGR contributors, 2021) and the *raster* (Hijmans, 2019) and *sf* R-packages (Pebesma, 2018).

2.5. Statistics

To test if CO_2 concentrations decreased from the first (upstream) to last (downstream) site along each stream, we used a Wilcoxon test for paired differences as differences were not normal distributed. We used linear regression models to examine the relationships between: (a) CO_2 concentration and chlorophyll *a* upstream and downstream of lakes, (b) CO_2 concentration, water temperature and chlorophyll *a* in River Guden, and (c) CO_2 emissions, groundwater index and catchment area. Non-normally distributed variables were log_{10} -transformed prior to analysis to meet model assumptions. Model selection was performed by stepwise backwards elimination using *F*-tests. Quantile regression models were applied to examine the relationship between the CO_2 concentration and emission distributions and predictors using the *quantreg* R-package (Koenker, 2021). When quantile regression models were not significant we applied locally estimated scatterplot smoothing to visualize relationships. All statistics were done using statistical programming software R (R Core Team, 2021).

3. Results

3.1. Spatial Patterns in CO₂ Concentration Across Streams

Due to substantial variation of CO₂ concentrations between individual streams, combining all CO₂ measurements showed a wide scatter though with an overall decline from upstream sites to downstream sites described as a function of the wetted area (Figure 2). Three quantile regression models (0.1-, 0.5-, and 0.9-quantiles) were all highly significant (p < 0.001) showing that the upper-, median- and lower bounds of the $log_{10}(CO_2)$ declined downstream in fluvial networks without lakes.





Figure 2. CO_2 concentrations during summer at many sites with distant or no influence of lakes (open circles) and below lakes (closed circles) along the stream course. Solid lines are quantile regression models (0.1-, 0.5-, and 0.9-quantiles, for all parameter estimates p < 0.001). Stream sites with lake influence (solid point) are not included in the regression analysis.

In 168 sites from 35 Danish streams with no lake influence, CO_2 concentrations on summer mornings were markedly supersaturated (mean \pm 95% CI: 159 \pm 15 μ M; 9.2-fold supersaturated). In 22 of these streams (with 4–10 sites distributed along the main stream course), CO_2 concentrations decreased (Wilcoxon test, V = 271, p < 0.001) downstream, while only one stream (River Vindinge) that was artificially widened at its upper site tended to have increasing CO_2 concentrations downstream (Table S2 in Supporting Information S1).

The concentrations of CO₂ measured in proximity of the inlet were markedly higher than those in the stream waters at the outlet of lakes (Figure 3). Lower CO₂ concentrations were measured both in lake outlets and in the following downstream sites (River Mølle and River Guden; Figure 3). In these streams with one or more lakes, mean CO₂ concentrations measured in the morning were supersaturated upstream of the lakes (152.1 μ M), but close to air saturation in lake outlets (16.8 μ M; Table S4 in Supporting Information S1). With higher phytoplankton chlorophyll *a* concentrations in the lakes, CO₂ concentrations were markedly lower in lake outlets (Figure 4; linear regression model reported as parameter estimate (±SE), $\log_{10}(CO_2) = 2.0(\pm 0.31) - 0.9(\pm 0.21) \cdot \log_{10}$ (chlorophyll *a*), $R^2 = 0.47$, N = 24, p < 0.001).

River Guden is a large network of streams and lakes with 10 tributaries and several lakes in a 2,649 km² catchment. Increasing water temperatures and chlorophyll *a* from upstream to downstream sites were strong predictors of decreasing CO₂ concentrations (Figure 5). A regression model of measurements of CO₂ at 61 sites in the network of streams and lakes showed a strong significant negative influence of both higher chlorophyll *a* and water temperature $(\log_{10}(CO_2) = 2.99(\pm 0.134) - 0.055(\pm 0.011) \cdot$ water temperature $-0.523(\pm 0.065) \cdot \log_{10}$ (chlorophyll *a*), $R^2 = 0.72$, N = 122, interaction not significant).

3.2. Seasonal Variation in CO₂ Concentrations

Four neighboring streams on Zealand were sampled biweekly for a full year (Figure 6). Mean CO_2 concentrations in outlets from four phytoplankton-rich lakes were, on average, 3-fold lower during mid-summer (June–August, mean 28 μ M, range 2–62 μ M) compared to the rest of the year (mean 91 μ M, range 31–139 μ M). Fourteen sites with no lake influence in the same streams had much higher mean CO_2 concentrations that were only 1.2-fold lower during mid-summer (mean 220, range 160–345 μ M) than the rest of the year (mean 266, range 196–422 μ M).

The biweekly CO_2 concentrations are shown for the River Pøle (Figure 7). CO_2 concentrations were high at upstream sites 1 and 2 with few plants (annual means 335 and 286 μ M), while concentrations were significantly





Figure 3. CO_2 concentrations on summer mornings along two streams without lakes (River Græse and River Havelse) and two streams with lakes (River Mølle and River Guden, tributaries omitted). The dotted line is the approximate CO_2 concentration at air saturation.



Figure 4. CO_2 concentrations in outlets from 12 lakes measured in the morning (8:00–11:00, open circles) and later in the afternoon (14:00–17:00, closed circles) as a function of the phytoplankton biomass (as chlorophyll *a*) in the lakes. Solid line is a linear regression model with 95% CI shown in gray. The *x*- and *y*-axis are log_{10} -transformed.





Figure 5. 3-D plot showing morning $\log_{10}(CO_2)$ at 61 sites in the River Guden stream-lake network as a function of water temperature and $\log_{10}(chlorophyll a)$. The plane is the fitted regression model of $\log_{10}(CO_2)$ as a function of $\log_{10}(chlorophyll a)$ and water temperature ($n = 61, R^2 = 0.75$).



Figure 6. CO_2 concentrations influenced by season and the presence or lack of lakes. Mean CO_2 concentrations at sites in four neighboring streams located in northern Zealand measured in the summer (June–August, *x*-axis) versus the rest of the year (September–May, *y*-axis). Four of the sites were located downstream of lakes (closed circles) and 14 sites were not influenced by lakes (open circles). The red diamond is the approximate CO_2 concentration at air saturation. The three lines represent CO_2 concentrations in summer (*x*) compared to the rest of the year (*y*) of 1:1, 1:2 and 1:3.

lower and seasonally variable at the mid-stream site 3, which contains many submerged plants only in summer (annual mean \pm 95% CI; 106 \pm 30 μ M). At the deeper downstream site 4 with no plants, CO₂ was higher and less variable seasonally (169 \pm 26 μ M). The River Pøle flows into the hypertrophic Lake Arre (site 5), which showed very low CO₂ concentrations (<1 μ M) below air saturation during mid-summer and appreciably higher mean CO₂ concentrations above air saturation (mean 58 μ M), during winter. The outlet from Lake Arre (site 6) had CO₂ concentrations closely resembling those in the middle of the lake basin (Figure 7).

The much stronger lowering of CO_2 concentrations downstream of lakes during summer than the rest of the year was not restricted to Lake Arre. The same seasonal pattern was found in the other three lakes on Zealand that were sampled biweekly over the year (seasonal means in Figure 6, biweekly data not shown).

3.3. CO₂ Emission

The CO₂ emissions were proportional to CO₂ concentrations (Figure S1 in Supporting Information S1) and varied extensively but tended to be highest at up- and mid-stream sites and lowest at downstream sites (Figure 8a) in accordance with the falling CO₂ concentrations downstream (Figure 2). The upper emission rates decreased with water temperatures between 8 and 24°C, but rates were highly variable close to 8°C, and less variable at water temperatures close to 20°C (Figure 8b). Downstream of phytoplankton-rich lakes, CO₂ fluxes were particularly low across the whole temperature range and often directed from air to water due to CO₂ undersaturation (Figure 8).

The mean daily CO₂ emissions from stream sites with no or distant lake influence ranged ten-fold, from 0.41 to 4.06 g C m⁻² (Table S1 in Supporting Information S1). Emissions were systematically higher in streams with high input of CO₂-rich groundwater. Thus, for 32 streams for which the groundwater index could be calculated, a linear regression model showed a significant positive relationship of CO₂ emission to the groundwater index (p = 0.02) and an almost significant negative relationship to the catchment area (\log_{10} -transformed, p = 0.08; CO₂ emission = 2.09 (±1.12) + 2.72(±1.13) · groundwater index – 0.78(±0.43) · \log_{10} (catchment area), $R^2 = 0.18$, N = 32; Figure S2 in Supporting Information S1).

4. Data Interpretation and Discussion

4.1. Spatiotemporal Variation of Stream CO₂ Concentrations

Generally, CO_2 concentrations in streams are highly influenced by the CO_2 supply from land and the atmospheric loss (Marx et al., 2017; Wallin et al., 2013). All Danish streams are short and there are no sharp distinction between steep headwater sites and river valley sites as in very large rivers deriving from inland mountains. The two Danish streams, Funder and Salten, with most headwater sites and steep downstream slope have a high groundwater index (0.85–0.91) and a high daily loss of CO_2 to the atmosphere (3.9–4.1 g C m⁻²), while two streams, Odense and Ryå with headwater sites of lower slopes, have lower groundwater indices (0.51–0.59) and CO_2 emissions (0.8–1.2 g C m⁻²; Table S2 in Supporting Information S1). Further, headwaters receiving much CO_2 -rich groundwater temperature of 8°C had the highest CO_2 levels, a characteristic also observed in other





Figure 7. Seasonal changes of CO_2 concentrations at 6 sites located from upstream to downstream in the River Pøle-Lake Arre network. Sites 1 and 2 are located in the upper river; site 3 is downstream a shallow reach with fast flow and dense vegetation; site 4 is the deep inlet to Lake Arre, devoid of vegetation; site 5 is in the middle of hypertrophic Lake Arre; site 6 is the outlet from Lake Arre. Measurements taken twice monthly from Aug 1995 to Aug 1996. The dotted line is the approximate CO_2 concentration at air saturation. Lakes and streams shown in inset (OpenStreetMap contributors, 2021).

European groundwater-fed alkaline streams (Demars & Trémolières, 2009; Maberly et al., 2015). This effect was reflected in falling CO_2 concentrations during mid-summer in the River Guden's fluvial network as water temperature gradually increased above the groundwater temperature. The falling CO_2 concentrations with increasing water temperature from 8 to 22°C reflected the more downstream location of warmer sites and longer water retention time in the networks, leaving more time for dissolved CO_2 in the stream water to equilibrate with the atmosphere or to be consumed by photosynthesis.

The observed downstream decline of CO_2 concentrations was not homogeneous and constant in all observed streams, but were influenced by the density of vegetation and the atmospheric loss. Transit through shallow, fast-flowing reaches with high gas transfer velocity and CO_2 emission (Equation 1) or through reaches with dense plant stands and their associated high CO_2 consumption in photosynthesis, may lead to local minimum CO_2 concentrations (e.g., site 3 in River Pøle; Figure 7). In deeper downstream reaches (e.g., site 4 in River Pøle), CO_2 concentrations may have increased due to slower flow, 4-fold lower CO_2 transfer velocity to the atmosphere and fewer plants compared with site 3. Very marked changes were observed during the transit through lakes, bringing several-fold CO_2 supersaturation upstream of a lake all the way to below air saturation downstream from the lake. Thus, in Danish stream-lake networks (e.g., River Guden, River Mølle, and River Pøle-Lake Arre; Figures 3 and 7) as well as reported patterns from streams in England and France (Maberly et al., 2015; Neal et al., 1998), CO_2 concentrations show high spatial variability.





Figure 8. Predictors of stream CO_2 emission. (a) CO_2 flux from stream sites located either downstream of lakes (closed circles) or with no or distant influence from lakes (open circles) as a function of the wetted area. The trend in the point cloud is shown using a locally estimated scatterplot smoothing (LOESS) function (solid line) with 95% CI in gray. The *x*-axis is log_{10} -transformed. (b) CO_2 flux from stream sites as a function of water temperature. The trend in the point cloud is shown using a LOESS smoothing function (solid line) with 95% CI in gray.

When CO_2 concentrations from all examined sites in Danish streams of widely different hydrology, geomorphology, and plant density were related to the wetted area across the stream, the spatial variability of CO_2 accounted for was low, though the overall downstream decrease was highly significant. Concurrently, CO_2 concentration also varied extensively between pronounced supersaturation at most stream sites and uptake in a few lake outlets. The CO_2 concentrations were low in lake outlets and even lower at sites several kilometers downstream. The CO_2 concentrations in lake outlets is reduced most during summer at high phytoplankton production and long water retention time, but even during winter at low phytoplankton production, CO_2 concentrations are much lower downstream than upstream as CO_2 continues to be emitted from the large lake surfaces (Figures 6 and 7). CO_2 concentrations decreased in lake outlets log-log linearly with higher phytoplankton biomass and were markedly undersaturated (only 1 µM) in the highly eutrophic Lake Arre (Figure 7). This is a result of intensive summer phytoplankton production, which changes to CO_2 supersaturation for the same lakes between autumn and spring due to less incoming light, limited photosynthesis and net heterotrophy (Figure 6). The seasonal variations of CO_2 concentrations are most pronounced in outlets from eutrophic lakes such as Lake Arre with a 100-fold variation from mid-summer (1 µM) to mid-winter (100 µM). In comparison, humic forest lakes remain CO_2 supersaturated and net heterotrophic during the entire year due to combined light and nutrient limitation of phytoplankton production (Martinsen et al., 2020b; Staehr et al., 2010). While, no fluvial networks with forest lakes were included in our study of the primarily open, cultivated Danish landscape, such lakes are common over large forested areas in boreal regions (Nydahl et al., 2020; Sobek et al., 2003). Thus, high spatial variability along streams and localized influence of groundwater inflow, land use and lakes must be accounted for through a high spatial coverage when CO_2 emission rates are up-scaled to emissions budgets for individual streams or large geographical regions (Duvert et al., 2018).

4.2. CO₂ Emission Rates

 CO_2 emission during summer varied between the shallow streams in the Danish landscape characterized by changing CO_2 input from land and in-stream respiration versus CO_2 loss to the atmosphere and consumption by photosynthesis of submerged plants and microalgae. The variability was particularly high in fluvial networks where CO_2 emission from streams upstream of lakes changed to CO_2 uptake downstream of lakes. However, shallow reaches with dense submerged plants can also support high summer gross photosynthetic rates (max. 2.8 g C m⁻² d⁻¹; Alnoee et al., 2021) effectively reducing CO_2 emission during the daytime. Subsequently, CO_2 emission increases during nocturnal respiration and later during autumn degradation of organic carbon accumulated in the plant biomass (Kelly et al., 1983; Sand-Jensen, 1998). Both on daily and annual basis the Danish streams are generally net heterotrophic according to oxygen mass balances with in-stream respiration exceeding gross photosynthesis and generating a CO_2 surplus (Alnoee et al., 2021; Kelly et al., 1983; Simonsen & Harremoës, 1978).

Although mean CO₂ emission from streams with no lakes varied 10 times from minimum to maximum, the overall summer mean for all streams was constrained within a relatively narrow confidence interval (1.75–2.50 g C m⁻² d⁻¹). Thus, it is possible to attain a reliable estimate of the mean value for Danish fluvial networks during summer, when, as in this study, several hundred sites are included representing the main geographical regions and fluvial networks (Gómez-Gerner, 2021). However, upscaling is challenged by the extreme spatial variability and the need to include numerous sites to account for the profound influence of: (a) lakes and their variable water retention time and phytoplankton productivity, (b) stream site location in the network and, thus, the variable downstream and lateral CO₂ input, and (c) variable in-stream autotrophic-heterotrophic balance.

We found that the mean summer emission rates per m^2 for Danish streams resembles global emissions for streams in temperate regions (Lauerwald et al., 2015). However, it remains uncertain how representative summer emissions are of annual emissions. Higher inflow of CO₂-rich soil water, including much drainage water from cultivated fields, during winter than summer as well as increasing respiration relative to photosynthesis in the fluvial network will generate a CO₂ surplus (Alnoee et al., 2021; Kelly et al., 1983; Simonsen & Harremoës, 1978). The increase of CO₂ concentrations and CO₂ emission from summer to winter is highest close to lake outlets and in lower reaches, while smaller changes are expected in groundwater-fed headwater streams with relatively constant CO₂ concentrations, water temperatures, and flow velocities (Sand-Jensen & Staehr, 2012).

It is important to determine annual CO_2 emissions in streams and validate the calculations by using different approaches. It is possible to combine continuously recording field sensors (e.g., flow velocity, depth, water temperature, CO_2 , pH, and conductivity) and gas flux chamber measurements (Sand-Jensen & Staehr, 2012), and compare them with general calculations based on regional monitoring data (i.e., discharge, slope, channel dimensions, etc.; Raymond et al., 2013). It should be possible to check these direct and general calculations with a mass balance approach using radon as a tracer of groundwater inflow and gas emission from stream surfaces (Duvert et al., 2019).

While attempting to evaluate patterns of CO_2 concentration and emission in streams at regional and global scales, we must recall that availability of CO_2 at local stream sites also has a direct influence on species diversity

23335084, 2022, 12, Downk

and primary production of plants that, in turn, affect the flow, water chemistry and CO_2 emission (Demars & Trémolières, 2009; Sand-Jensen, 1998). Carbon dynamics remains a complexity of regulations and interactions at local and catchment levels in fluvial networks.

Data Availability Statement

Data on stream discharge and the national elevation model are publicly available from the sources cited in the main text. Data and scripts used for the analysis and figures are available from an online repository (https://doi. org/10.17894/ucph.108eeee5-131c-4eab-bc5a-0a338d729e77; Sand-Jensen, Riis, Kjær, & Martinsen, 2022).

References

- Abril, G., Bouillon, S., Darchambeau, F., Teodoru, C. R., Marwick, T. R., Tamooh, F., et al. (2015). Technical Note: Large overestimation of pCO₂ calculated from pH and alkalinity in acidic, organic-rich freshwaters. *Biogeosciences*, 12(1), 67–78. https://doi.org/10.5194/bg-12-67-2015
- Alnoee, A. B., Levi, P. S., Baattrup-Pedersen, A., & Riis, T. (2021). Macrophytes enhance reach-scale metabolism on a daily, seasonal and annual basis in agricultural lowland streams. Aquatic Sciences, 83(1), 11. https://doi.org/10.1007/s00027-020-00766-4
- Barnes, R. (2018). RichDEM: High-performance terrain analysis. PeerJ, e27099v1. Preprint.
- Barnes, R., Lehman, C., & Mulla, D. (2014). An efficient assignment of drainage direction over flat surfaces in raster digital elevation models. Computers & Geosciences, 62, 128–135. https://doi.org/10.1016/j.cageo.2013.01.009
- Berner, E. K., & Berner, R. A. (2012). Global environment: Water, air, and geochemical cycles. Princeton University Press.
- Borges, A. V., Darchambeau, F., Lambert, T., Bouillon, S., Morana, C., Brouyère, S., et al. (2018). Effects of agricultural land use on fluvial carbon dioxide, methane and nitrous oxide concentrations in a large European river, the Meuse (Belgium). Science of the Total Environment, 610–611, 342–355. https://doi.org/10.1016/j.scitotenv.2017.08.047
- Brinkerhoff, C. B., Raymond, P. A., Maavara, T., Ishitsuka, Y., Aho, K. S., & Gleason, C. J. (2021). Lake morphometry and river network controls on evasion of terrestrially sourced headwater CO₂. Geophysical Research Letters, 48(1), e2020GL090068. https://doi.org/10.1029/2020GL090068
- Cole, J. J., Prairie, Y. T., Caraco, N. F., McDowell, W. H., Tranvik, L. J., Striegl, R. G., et al. (2007). Plumbing the global carbon cycle: Integrating inland waters into the terrestrial carbon budget. *Ecosystems*, 10(1), 172–185. https://doi.org/10.1007/s10021-006-9013-8
- Crawford, J. T., Loken, L. C., Stanley, E. H., Stets, E. G., Dornblaser, M. M., & Striegl, R. G. (2016). Basin scale controls on CO₂ and CH₄ emissions from the Upper Mississippi River. *Geophysical Research Letters*, 43(5), 1973–1979. https://doi.org/10.1002/2015GL067599
- Demars, B. O. L., & Trémolières, M. (2009). Aquatic macrophytes as bioindicators of carbon dioxide in groundwater fed rivers. Science of the Total Environment, 407(16), 4752–4763. https://doi.org/10.1016/j.scitotenv.2009.04.017
- Drake, T. W., Raymond, P. A., & Spencer, R. G. M. (2018). Terrestrial carbon inputs to inland waters: A current synthesis of estimates and uncertainty. *Limnology and Oceanography Letters*, 3, 132–142. https://doi.org/10.1002/lol2.10055
- Duvert, C., Bossa, M., Tyler, K. J., Wynn, J. G., Munksgaard, N. C., Bird, M. I., et al. (2019). Groundwater-derived DIC and carbonate buffering enhance fluvial CO₂ evasion in two Australian tropical rivers. *Journal of Geophysical Research: Biogeosciences*, 124(2), 312–327. https://doi. org/10.1029/2018jg004912
- Duvert, C., Butman, D. E., Marx, A., Ribolzi, O., & Hutley, L. B. (2018). CO₂ evasion along streams driven by groundwater inputs and geomorphic controls. *Nature Geoscience*, 11, 813–818. https://doi.org/10.1038/s41561-018-0245-y
- Finlay, J. C. (2003). Controls of streamwater dissolved inorganic carbon dynamics in a forested watershed. *Biogeochemistry*, 62(3), 231–252. https://doi.org/10.1023/A:1021183023963
- Fuß, T., Behounek, B., Ulseth, A. J., & Singer, G. A. (2017). Land use controls stream ecosystem metabolism by shifting dissolved organic matter and nutrient regimes. *Freshwater Biology*, 62(3), 582–599. https://doi.org/10.1111/fwb.12887
- GDAL/OGR contributors. (2021). GDAL/OGR geospatial data abstraction software library. Open Source Geospatial Foundation.
- Gómez-Gener, L., Rocher-Ros, G., Battin, T., Cohen, M. J., Dalmagro, H. J., Dinsmore, K. J., et al. (2021). Global carbon dioxide efflux from rivers enhanced by high nocturnal emissions. *Geoscience, Nature Geoscience*, 14(5), 5–294. https://doi.org/10.1038/s41561-021-00722-3
- Gran, G. (1952). Determination of the equivalence point in potentiometric titrations. Part II. Analyst, 77(920), 661–671. https://doi.org/10.1039/ an9527700661
- Hijmans, R. J. (2019). raster: Geographic data analysis and modeling. Retrieved from https://CRAN.R-project.org/package=raster
- Hotchkiss, E. R., Hall, R. O., Jr., Sponseller, R. A., Butman, D., Klaminder, J., Laudon, H., et al. (2015). Sources of and processes controlling CO₂ emissions change with the size of streams and rivers. *Nature Geoscience*, 8(9), 696–699. https://doi.org/10.1038/ngeo2507
- Hutchins, R. H. S., Prairie, Y. T., & Giorgio, P. A. d. (2019). Large-scale landscape drivers of CO₂, CH₄, DOC, and DIC in boreal river networks. *Global Biogeochemical Cycles*, 33(2), 125–142. https://doi.org/10.1029/2018GB006106
- Jespersen, A.-M., & Christoffersen, K. (1987). Measurements of chlorophyll——a from phytoplankton using ethanol as extraction solvent. Archiv für Hydrobiologie, 109(3), 445–454.
- Johnson, M. S., Lehmann, J., Riha, S. J., Krusche, A. V., Richey, J. E., Ometto, J. P. H. B., & Couto, E. G. (2008). CO₂ efflux from Amazonian headwater streams represents a significant fate for deep soil respiration. *Geophysical Research Letters*, 35(17), L17401. https://doi. org/10.1029/2008gl034619
- Kelly, M. G., Thyssen, N., & Moeslund, B. (1983). Light and the annual variation of oxygen-based measurements og productivity in a macrophyte-dominated river. *Limnology & Oceanography*, 28(3), 503–515. https://doi.org/10.4319/10.1983.28.3.0503
- Koenker, R. (2021). quantreg: Quantile regression. In (Version R package version 5.86). Retrieved from https://CRAN.R-project.org/package=quantreg
- Lauerwald, R., Laruelle, G. G., Hartmann, J., Ciais, P., & Regnier, P. A. G. (2015). Spatial patterns in CO₂ evasion from the global river network. *Global Biogeochemical Cycles*, 29(5), 534–554. https://doi.org/10.1002/2014GB004941
- Lindsay, J. B. (2016). Efficient hybrid breaching-filling sink removal methods for flow path enforcement in digital elevation models. *Hydrological Processes*, 30(6), 846–857. https://doi.org/10.1002/hyp.10648
- Liu, S., & Raymond, P. A. (2018). Hydrologic controls on pCO₂ and CO₂ efflux in US streams and rivers. *Limnology and Oceanography Letters*, 3(6), 428–435. https://doi.org/10.1002/2Flo12.10095

Acknowledgments

We thank Independent Research Fund Denmark (0217-00112B) for a grant to KSJ and the project: "Supporting climate and biodiversity by rewetting low-lying areas." JEK and KTM were associated with this project. TR was supported by the VELUX Foundation. We thank David Stuligross for constructive comments and careful proofreading.

23335084, 2022, 12, Downk

- Maberly, S. C., Berthelot, S. A., Stott, A. W., & Gontero, B. (2015). Adaptation by macrophytes to inorganic carbon down a river with naturally variable concentrations of CO₂. *Journal of Plant Physiology*, *172*, 120–127. https://doi.org/10.1016/j.jplph.2014.07.025
- Martinsen, K. T., Kragh, T., & Sand-Jensen, K. (2020a). Carbon dioxide efflux and ecosystem metabolism of small forest lakes. Aquatic Sciences, 82, 1–17. https://doi.org/10.1007/s00027-019-0682-8
- Martinsen, K. T., Kragh, T., & Sand-Jensen, K. (2020b). Carbon dioxide partial pressure and emission throughout the Scandinavian stream network. Global Biogeochemical Cycles, 34(12), e2020GB006703. https://doi.org/10.1029/2020GB006703
- Marx, A., Dusek, J., Jankovec, J., Sanda, M., Vogel, T., Van Geldern, R., et al. (2017). A review of CO₂ and associated carbon dynamics in headwater streams: A global perspective. *Reviews of Geophysics*, 55(2), 560–585. https://doi.org/10.1002/2016rg000547
- Maurice, L., Rawlins, B. G., Farr, G., Bell, R., & Gooddy, D. C. (2017). The influence of flow and bed slope on gas transfer in steep streams and their implications for evasion of CO₂. Journal of Geophysical Research: Biogeosciences, 122(11), 2862–2875. https://doi. org/10.1002/2017jg004045
- MFM, & DCE. (2021). Surface water database. Ministry of Environment and Food of Denmark, Danish Centre for Environment and Energy. Retrieved from https://odaforalle.au.dk/
- Neal, C., House, W., Jarvie, H., & Eatherall, A. (1998). The significance of dissolved carbon dioxide in major lowland rivers entering the North Sea. Science of the Total Environment, 210–211, 187–203. https://doi.org/10.1016/s0048-9697(98)00012-6

Nydahl, A. C., Wallin, M. B., & Weyhenmeyer, G. A. (2020). Diverse drivers of long-term p CO₂ increases across thirteen boreal lakes and streams. *Inland Waters*, 10(3), 360–372. https://doi.org/10.1080/20442041.2020.1740549

O'Callaghan, J. F., & Mark, D. M. (1984). The extraction of drainage networks from digital elevation data. Computer Vision, Graphics, and Image Processing, 28(3), 323–344. https://doi.org/10.1016/S0734-189X(84)80011-0

OpenStreetMap contributors. (2021). OpenStreetMap. Retrieved from https://planet.osm.org

- Ovesen, N. B., Iversen, H. L., Larsen, S. E., Müller-Wohlfeil, D.-I., Svendsen, L. M., Blicher, A. S., & Jensen, P. M. (2000). Drainage conditions in Danish streams (in Danish with English summary and legends). *Danmarks Miljøundersøgelser (DMU)*. Report 340
- Pebesma, E. (2018). Simple features for R: Standardized support for spatial vector data. *The R Journal*, 10(1), 439–446. https://doi.org/10.32614/ RJ-2018-009
- Raymond, P. A., Hartmann, J., Lauerwald, R., Sobek, S., McDonald, C., Hoover, M., et al. (2013). Global carbon dioxide emissions from inland waters. *Nature*, 503(7476), 355–359. https://doi.org/10.1038/nature12760

R Core Team. (2021). R: A language and environment for statistical computing. Retrieved from https://www.R-project.org/

- Rebsdorf, A., Thyssen, N., & Erlandsen, M. (1991). Regional and temporal variation in pH, alkalinity and carbon dioxide in Danish streams, related to soil type and land use. *Freshwater Biology*, 25(3), 419–435. https://doi.org/10.1111/j.1365-2427.1991.tb01386.x
- Rubek, F., Scharling, M., & Cappelen, M. (2020). Denmark's climate (in Danish with English summary). Danmarks Meteorologiske Institut (DMI). DMI Report 21-01
- Sand-Jensen, K. (1998). Influence of submerged macrophytes on sediment composition and near-bed flow in lowland streams. Freshwater Biology, 39(4), 663–679. https://doi.org/10.1046/j.1365-2427.1998.00316.x
- Sand-Jensen, K., Friberg, N., & Murphy, J. (2006). Running waters. Historical development and restoration of lowland Danish streams. National Environmental Research Institute.

Sand-Jensen, K., & Frost-Christensen, H. (1998). Photosynthesis of amphibious and obligately submerged plants in CO₂-rich lowland streams. Oecologia, 117(1-2), 31-39. https://doi.org/10.2307/4222130

- Sand-Jensen, K., Riis, T., Kjær, J. E., & Martinsen, K. T. (2022). CO₂ lowland streams dataset [Dataset]. ERDA. https://doi.org/10.17894/ ucph.108eeee5-131c-4eab-bc5a-0a338d729e77
- Sand-Jensen, K., Riis, T., & Martinsen, K. T. (2022). Photosynthesis, growth, and distribution of plants in lowland streams—A synthesis and new data analyses of 40 years research. *Freshwater Biology*, 67(7), 1255–1271. https://doi.org/10.1111/fwb.13915
- Sand-Jensen, K., & Staehr, P. A. (2012). CO₂ dynamics along Danish lowland streams: Water-air gradients, piston velocities and evasion rates. *Biogeochemistry*, 111(1-3), 615–628. https://doi.org/10.1007/s10533-011-9696-6
- SDFE. (2021). Danish map supply, SDFE (Agency for Datasupply and Efficiency). Retrieved from https://download.kortforsyningen.dk/
- Simonsen, J. F., & Harremoës, P. (1978). Oxygen and pH fluctuations in rivers. Water Research, 12(7), 477-489. https://doi.org/10.1016/0043-1354(78)90155-0
- Smits, A. P., Schindler, D. E., Holtgrieve, G. W., Jankowski, K. J., & French, D. W. (2017). Watershed geomorphology interacts with precipitation to influence the magnitude and source of CO₂ emissions from Alaskan streams. *Journal of Geophysical Research: Biogeosciences*, 122(8), 1903–1921. https://doi.org/10.1002/2017jg003792
- Sobek, S., Algesten, G., Bergström, A. K., Jansson, M., & Tranvik, L. J. (2003). The catchment and climate regulation of pCO₂ in Boreal lakes. *Global Change Biology*, 9(4), 630–641. https://doi.org/10.1046/j.1365-2486.2003.00619.x
- Staehr, P. A., Sand-Jensen, K., Raun, A. L., Nilsson, B., & Kidmose, J. (2010). Drivers of metabolism and net heterotrophy in contrasting lakes. Limnology & Oceanography, 55(2), 817–830. https://doi.org/10.4319/lo.2010.55.2.0817
- Tarboton, D. G. (2017). Terrain analysis using digital elevation models (TauDEM) Utah Water Research Laboratory. Utah State University. Retrieved from http://hydrology.usu.edu/taudem/taudem5/index.html
- Thyssen, N., & Erlandsen, M. (1987). Reaeration of oxygen in shallow, macrophyte-rich streams: II. Relationship between the reaeration rate coefficient and hydraulic properties. *Internationale Revue der gesamten Hydrobiologie und Hydrographie*, 72(5), 575–597. https://doi.org/10.1002/iroh.19870720505
- Wallin, M. B., Grabs, T., Buffam, I., Laudon, H., Ågren, A., Öquist, M. G., & Bishop, K. (2013). Evasion of CO₂ from streams-The dominant component of the carbon export through the aquatic conduit in a boreal landscape. *Global Change Biology*, 19(3), 785–797. https://doi. org/10.1111/2Fgcb.12083
- Wallin, M. B., Öquist, M. G., Buffam, I., Billett, M. F., Nisell, J., & Bishop, K. H. (2011). Spatiotemporal variability of the gas transfer coefficient (KCO₂) in boreal streams: Implications for large scale estimates of CO₂ evasion. *Global Biogeochemical Cycles*, 25(3), GB3025. https://doi. org/10.1029/2F2010gb003975